cs330 - Discrete Structures, Fall 1995

Computer Science and Applied Mathematics

Programming Assignment 2 - Grading

Student:

Functionality pass fail Readability pass fail

32 bit SSN -0.2 Memo included -0.3 Correct DFA -0.5 Standard memo format -0.2 Correct number of SSN occurences -0.5 Proper file documentation -0.3 Correct O() for time. It ’s O(n) for most designs but it may be different for oth- ers -0.5 Proper function(s) documentation -0.3

Correct O() for space. It ’s O(1) for most designs but it may be different for others

-0.2

Specify the hardware platform used -0.2 Specify the software environment used -0.2 Extra credit for measuring the time complexity

+0.2

Functionality x Readability = Final mark